

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Bietenbeck, Jan

Working Paper Own Motivation, Peer Motivation, and Educational Success

IZA Discussion Papers, No. 13872

Provided in Cooperation with: IZA – Institute of Labor Economics

Suggested Citation: Bietenbeck, Jan (2020) : Own Motivation, Peer Motivation, and Educational Success, IZA Discussion Papers, No. 13872, Institute of Labor Economics (IZA), Bonn

This Version is available at: https://hdl.handle.net/10419/232624

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 13872

Own Motivation, Peer Motivation, and Educational Success

Jan Bietenbeck

NOVEMBER 2020



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 13872

Own Motivation, Peer Motivation, and Educational Success

Jan Bietenbeck Lund University, CESifo, DIW Berlin and IZA

NOVEMBER 2020

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9	Phone: +49-228-3894-0	
53113 Bonn, Germany	Email: publications@iza.org	www.iza.org

ABSTRACT

Own Motivation, Peer Motivation, and Educational Success*

I study how motivation shapes own and peers' educational success. Using data from Project STAR, I find that academic motivation in early elementary school, as measured by a standardized psychological test, predicts contemporaneous and future test scores, high school GPA, and college-test taking over and above cognitive skills. Exploiting random assignment of students to classes, I find that exposure to motivated classmates causally affects contemporaneous reading achievement, a peer effect that operates over and above spillovers from classmates' past achievement and socio-demographic composition. However, peer motivation does not affect longer-term educational success, likely because it does not change own motivation.

JEL Classification:	I21, J13, J24
Keywords:	motivation, personality, peer effects, Project STAF

Corresponding author:

Jan Bietenbeck Lund University Department of Economics P.O. Box 7080 22007 Lund Sweden E-mail: mail@janbietenbeck.com

^{*} Thanks to Fabian Kosse, Kaveh Majlesi, Therese Nilsson, Tove Nordström, Pia Pinger, Felix Weinhardt, Ulf Zölitz,

and participants of the 5th IZA Workshop on the Economics of Education for helpful comments.

1 Introduction

A growing literature in economics documents the importance of personality for individuals' educational and labor market success (Borghans et al., 2008; Almlund et al., 2011; Heckman, Jagelka, and Kautz, 2019). Much of this research has focused on preference parameters, such as patience and risk-taking, and personality traits, such as the ones captured by the widely-used Big Five taxonomy. In contrast, other facets of personality such as motivation, which features prominently in psychological research (e.g. Roberts, 2006), have so far received less attention by economists.¹ Moreover, despite extensive evidence that the social environment matters for performance in school and in the workplace (e.g. Mas and Moretti, 2009; Sacerdote, 2011), only very few studies have examined how individuals are affected by the personality of their peers.

In this paper, I extend the research on how personality shapes educational success by studying the role of motivation. I make two key contributions. First, I show that children's academic motivation, as measured by a standardized psychological test in early elementary school, predicts achievement in elementary and middle school, high school GPA, and the taking of a college entrance test around age 18 over and above cognitive skills. Second, I document spillovers of personality in the social environment: students who are exposed to more motivated classmates score higher on a standardized reading test in elementary school, a peer effect that operates over and above spillovers from classmates' past achievement and socio-demographic composition.

The empirical analysis uses data from the Tennessee Student-Teacher Achievement Ratio experiment (Project STAR), a study of class size effects that followed a single cohort of children who started kindergarten in one of 79 participating schools in 1985. These data are uniquely suited for my purposes for two reasons. First, the experiment measured students' academic motivation in grades 1 through 3 using a standardized psychological test, and the longitudinal data allow me to relate this measure to a host of short- and longer-term outcomes. Second, some children joined the study cohort at participating schools in second and third grade and were randomly assigned to classes upon entry. This generated exogenous, measurable variation in the (predetermined) motivation of their classmates, which I can use to estimate causal spillover effects.

In the first part of the paper, I show that children's academic motivation predicts their educational performance in the short and longer term. For example, a one standard deviation (SD) higher motivation during grades 1-3 is associated with 0.05-0.06 SD

¹ Psychologists distinguish between four core domains of personality (Roberts, 2006). Motivation is part of the motives domain and is notably distinct from traits, which form their own domain. I discuss models of personality in more detail in Section 2 below.

higher reading and math scores in elementary and middle school and a 4.4 percent higher likelihood to take a college entrance exam around age 18. These associations hold even when I control for reading and math achievement in kindergarten, suggesting that they are not due to correlated academic ability. Motivation is also associated with good classroom behavior, as rated by teachers, in fourth and eighth grade, a measure which previous studies have shown to predict educational attainment and earnings among participants in Project STAR (Chetty et al., 2011; Bietenbeck, 2020).

In the second part of the paper, I examine how children's motivation affects the learning outcomes of their peers. For this analysis, I focus on a sample of students who first entered the schools participating in Project STAR in second or third grade. These students were randomly assigned to classes within school upon entry, which allows me to avoid the selection problems that typically complicate the identification of causal peer effects. Moreover, the new classmates of these entrants had participated in the experiment in the previous school year, which lets me observe their predetermined motivation. My regressions exploit the variation in classmates' motivation due to the random assignment of entrants to classes in order to identify spillover effects.

The results show that peer personality matters for achievement in school. Specifically, a 1 SD increase in classmates' average motivation raises performance on a standardized reading test at the end of the school year by 0.07 SD. This estimate is robust to controlling for classmates' past achievement and socio-demographic background, which suggests that it reflects a true personality spillover. However, peer motivation does not appear to matter beyond contemporaneous achievement, as it does not affect any of the longer-term outcomes measured after the experiment ended and classes were reorganized at the end of third grade. Interestingly, peer motivation also does not affect contemporaneous own motivation.

Which mechanism explains this pattern of results? I argue that the spillover on contemporaneous achievement is most likely due to an improved learning environment in school, as motivated peers show better classroom behavior and distract their classmates less. As for the lack of longer-term effects, previous research has found that childhood interventions are particularly successful at changing future outcomes if they affect children's personality (e.g. Heckman, Pinto, and Savelyev, 2013). This suggests that the absence of longer-term impacts is due to the fact that peer motivation does not change own motivation. Put differently, it appears that the contemporaneous effect on reading scores by itself is simply not large enough to generate measurable longer-term impacts. I briefly discuss the implications of these findings in the conclusion.

This paper contributes to a large literature in economics and psychology that stud-

ies the importance of personality for success in life (for surveys, see Borghans et al., 2008; Almlund et al., 2011; Heckman, Jagelka, and Kautz, 2019). One strand of this literature documents the predictive power of different facets of personality for educational outcomes. This research has found that preference parameters, such as patience (e.g. Golsteyn, Grönqvist, and Lindahl, 2014; Cadena and Keys, 2015), and personality traits, such as conscientiousness (e.g. Poropat, 2009; Gensowski, 2018), grit (e.g. Duckworth et al., 2007), and locus of control (e.g. Piatek and Pinger, 2016), matter for human capital accumulation. Moreover, studies in psychology have shown that academic motivation predicts achievement (e.g. Wong and Csikszentmihalyi, 1991; Steinmayr and Spinath, 2009), but typically in small samples and with short follow-up periods. In economics, Heckman, Pinto, and Savelyev (2013) show that the Perry Preschool program improved treated children's achievement partly because it boosted their motivation.² I complement this research by documenting how motivation relates to short- and longer-term achievement, classroom behavior, and attainment in a large sample of children.

This paper also contributes to a large literature on peer effects in education (for surveys, see Sacerdote, 2011; Paloyo, 2020). This research, which includes papers using data from Project STAR, has mostly studied spillovers from peers' academic ability (e.g. Lavy, Silva, and Weinhardt, 2012; Sojourner, 2013; Booij, Leuven, and Oosterbeek, 2017; Feld and Zölitz, 2017; Bietenbeck, 2020) and socio-demographic composition (e.g. Hoxby, 2000; Whitmore, 2005; Lavy and Schlosser, 2011; Brenoe and Zölitz, 2019).

Two recent studies explicitly examine spillovers from personality and are thus closely related to this paper. First, Ballis (2020) studies a policy-driven change in the returns to schooling for undocumented youths in the United States. She shows that U.S.born high school peers of these youths, who did not benefit from the policy directly, performed better in school after its implementation. She interprets this effect as a spillover from undocumented youths' increased motivation, but she cannot measure motivation in her data. Second, Golsteyn, Non, and Zölitz (2020) exploit detailed data on personality in a university setting to show that students perform better in the presence of persistent peers. These effects operate over and above spillovers from academic ability and socio-demographic characteristics, leading the authors to conclude that they reflect true personality spillovers. I contribute to this research by studying spillovers from motivation in elementary school. Unlike the two previous studies, I am able to estimate relatively long-term effects. Moreover, I can study how peer motivation affects own motivation; to the best of my knowledge, mine are the first estimates of

² A separate literature in economics studies how extrinsic incentives can be used to boost motivation and school performance. See Koch, Nafziger, and Nielsen (2015) for an overview of this research.

spillovers from peer personality on own personality in the literature.³

The remainder of this paper is organized as follows. Section 2 summarizes the research on personality, and especially motivation, in psychology. Section 3 presents details on Project STAR and the data. In Section 4, I document how own motivation in early elementary school relates to short- and longer-term educational success. Section 5 presents estimates of spillovers from motivated peers. Section 6 concludes.

2 Motivation in personality psychology

Psychologists have developed a lot of different models of personality. The theoretical framework by Roberts (2006), which has been popularized in economics by Almlund et al. (2011), captures the key features of many of these models. According to this framework, the core of personality is made up by four domains: traits, motives, abilities, and narratives. Traits capture the relatively stable patterns of thoughts, feelings, and behaviors of an individual and are often represented using the well-known Big Five taxonomy.⁴ Motives are defined as what an individual desires, needs, and strives for. Abilities capture things such as intelligence, and narratives are the stories that an individual tells herself in order to make sense of her life. Together, these four domains shape a person's identity and reputation, which in turn determine her roles in society.

This paper studies the importance of academic motivation, which falls under the motives domain. Unlike the literature on personality traits, psychological research on motivation has not converged on a common theoretical framework, system of measurement, or terminology (Murphy and Alexander, 2000; Roberts et al., 2006). Despite this heterogeneity, empirical studies have consistently found that motivation is predictive of success in life: for example, Steinmayr and Spinath (2009) document that motivation predicts school performance over and above intelligence, and Dunifon and Duncan (1998) show that having an orientation toward challenge predicts future earnings. In

³ In related unpublished work, Shure (2017) uses data on secondary-school students in Belgium and shows that students with more conscientious classmates perform better in school. Her results rely on the relatively strong assumption that conditional on controls and school fixed effects, assignment to classes is as good as random. Some other papers do not explicitly study spillovers from personality but examine impacts of peers who likely exhibit disruptive behavior, such as children exposed to domestic violence (Carrell and Hoekstra, 2010; Carrell, Hoekstra, and Kuka, 2018) and boys with female-sounding names (Figlio, 2007). In Bietenbeck (2020), I study spillovers from low-achieving kindergarten repeaters in Project STAR and argue that they may arise due to misbehavior. However, the data do not allow me to measure repeaters' behavior or personality at baseline.

⁴ The Big Five traits are openness to experience, conscientiousness, extraversion, agreeableness, and neuroticism. Almlund et al. (2011) give an overview of different taxonomies of personality traits and their relation to the widely-studied concepts of grit and locus of control.

related work in economics, Segal (2012) finds that intrinsic motivation in adolescence and early adulthood, as measured by performance on a low-stakes coding speed test, predicts future earnings over and above cognitive skills.⁵

The apparent importance of motivation for success in life has led psychologists to study potential ways to boost motivation among students. Results show that interventions that directly aim at increasing motivation, for example by helping students set goals or by instructing teachers to relate lesson content to students' experiences, can improve motivation and achievement (see Lazowski and Hulleman, 2016, for a review of this evidence). Within economics, Heckman, Pinto, and Savelyev (2013) similarly show that the Perry Preschool intervention raised children's academic motivation. In contrast, previous analyses of Project STAR did not find any evidence that class size affects motivation (Word et al., 1990; Schanzenbach, 2006).

3 Project STAR: background and data

3.1 Background on Project STAR

Project STAR was a randomized controlled trial designed to investigate the effect of class size on student achievement. The original experiment followed a single cohort of children at 79 schools in Tennessee from kindergarten through third grade. It started at the beginning of the 1985-86 school year, when 6,325 kindergarten students were randomly assigned to small classes (target size 13-17 students) or regular-sized classes (target size 22-25 students) within their school.⁶ Because kindergarten was not mandatory at that time and due to normal residential mobility, 5,276 additional students joined this study cohort at participating schools during grades 1-3. These students were also randomized to classes within school upon entry, implying that the randomization pool for all participants was school-by-entry-grade. After the initial randomization, all students were supposed to stay in their assigned class type (small versus regular-sized) until the end of third grade, at which point the experiment ended. At the start of each

⁵ Only very few studies have examined potential correlations between motives and personality traits (Roberts et al., 2006), and this literature has yielded mixed results. Thus, Winter et al. (1998) find that extraversion is unrelated to two different measures of motivation among female college graduates. In contrast, Komarraju and Karau (2005) document a complex pattern of relationships between Big Five traits and several motivational constructs in a sample of undergraduate students, with a particularly strong correlation between conscientiousness and motivation to achieve.

⁶ There was also a third type of class: regular-sized class with a full-time teacher's aide. Previous studies using data from Project STAR have not found any differences in treatment effects between regular-sized classes with and without a full-time teacher's aide. In the empirical analysis, I follow the convention in the literature and group these two types of classes together.

grade, teachers were also randomly assigned to classes within school.

As with any field experiment, the actual implementation of Project STAR deviated somewhat from the original plan. Thus, as children advanced from kindergarten to third grade, some students managed to move between small and regular-sized classes (for details, see Krueger, 1999). To account for this likely non-random sorting, I always define peer composition based on the initial random assignment when I estimate spillovers from motivated classmates below. Another deviation from the original study design was that a substantial number of students left the experiment either because they moved to other schools or because they were retained in grade. Later on, I provide evidence that this attrition is not driving my results.⁷

3.2 Data and variable definitions

An important feature of Project STAR is that researchers collected detailed data on participants both during the experiment and long after it ended. Most of these data are included in the Project STAR public use file, which forms the basis for my empirical analysis and which allows me to follow students from kindergarten through the end of high school. In this Subsection, I give a brief overview of the main variables I draw from this dataset, with additional details provided in Online Appendix A.

Academic motivation. In the spring of each year from kindergarten through third grade, students' academic motivation was assessed using the Self-Concept and Motivation Inventory (SCAMIN; Milchus, Farrah, and Reitz, 1968). This is a groupadministered, standardized psychological test in which students are asked to indicate pictorially their response to different situations. Specifically, students are given a prepared answer sheet that contains a number of faces ranging from sad to happy for each situation. The test administrator – in Project STAR, this was the class teacher – then reads out a series of questions starting with "What face would you wear..." and asks students to mark the appropriate face as a response. For example, students are asked "What face would you wear if you were able to read like a grown-up?" and "What face would you wear if you could make the teacher happy with your arithmetic?"⁸ A motivation score is then calculated for each student based on her answers. This score serves as the measure of academic motivation in the empirical analysis below.

⁷ For additional details on the design and implementation of Project STAR, see Word et al. (1990), Krueger (1999), and Finn et al. (2007).

⁸ Unfortunately, the SCAMIN is out of print at the time of writing. From my reading of the secondary literature, its questions aim to measure both subject-specific motivation in reading, writing, and math, and more general motivation to achieve in school.

Besides academic motivation, the SCAMIN also measures students' academic selfconcept using a separate set of questions. Psychologists define self-concept as a person's perception of herself, which is formed through experience with her environment (Shavelson, Hubner, and Stanton, 1976). In the theoretical framework by Roberts (2006), self-concept forms part of a person's identity, which is shaped by the four core personality domains but which may itself also influence these domains via feedback processes. While self-concept is not the focus of this paper, I show in robustness checks that it does not confound my estimates for own and peer motivation.

As is usual for standardized tests for children, the SCAMIN has different test forms that are aimed at different grade levels: preschool/kindergarten, early elementary school, and late elementary school. In Project STAR, the preschool/kindergarten form was administered at the end of kindergarten and the early elementary form was administered at the end of grades 1-3. These forms differ in the questions that are asked and the number of faces that are shown on the answer sheet, such that motivation scores are not directly comparable between them.

Tests in personality psychology are often judged on various dimensions of quality, such as reliability and the ability to predict contemporaneous and future outcomes. As discussed in detail in Online Appendix A, the existing evidence points to a high quality of the SCAMIN early elementary form: for example, its test-retest reliability is similar to that found for tests measuring personality traits in children, and my results below show that its motivation score predicts a wide range of contemporaneous and future outcomes. Unfortunately, however, the preschool/kindergarten form does not meet this same high quality standard. In particular, there is some doubt about whether its questions capture only motivation, and I found in separate analyses that its motivation as captured by the early elementary form (see Online Appendix A). Given these serious problems, I decided not to use the kindergarten motivation scores and to focus only on motivation in grades 1-3 as measured by the SCAMIN early elementary form.

Achievement in reading and math. At the end of each grade from kindergarten through third grade, participants in Project STAR wrote the grade-appropriate version of the Stanford Achievement Test. Moreover, in the spring of grades 5-8, all students who were enrolled in public schools in Tennessee wrote the Comprehensive Test of Basic Skills as part of a statewide testing program. Both tests are standardized assessments covering various subjects, and I use the reading and math scores included in the Project STAR public use file as my main measures of student achievement. Classroom behavior. When STAR participants were in fourth grade, their teachers rated a subset of them on their classroom behavior. Teacher ratings for 28 behaviors were recorded on a scale from 1-5 and then consolidated into four indices. The effort index measures behaviors such as showing persistence when confronted with difficult problems. The initiative index captures things such as actively participating in classroom discussions. The discipline index measures behaviors such as being quiet versus interfering with classmates' work. The value index captures to what extent a student appreciates the school learning environment. All indices are coded such that higher values reflect better behavior. In eighth grade, math and English teachers rated a different subset of STAR participants using a similar but shorter questionnaire, and the ratings were consolidated into the same four indices. In the analysis below, I measure classroom behavior using the total of eight fourth- and eighth-grade indices.

Educational attainment. Most participants in Project STAR graduated from high school in 1998, and researchers collected information on the high school grade point average (GPA) and graduation status for participants attending selected high schools in 1999 and 2000. Besides this information, the public use file contains an indicator for whether a student had taken an ACT or SAT college-entrance test by 1998. This indicator is based on the administrative records of the two companies offering these tests and is the outcome of a data collection effort by Krueger and Whitmore (2001). It is available for the full sample of STAR participants and is a measure of college intent.

Student characteristics. The data contain information on the following socio-demographic characteristics of students: age, gender, race, and an indicator for whether the student was ever eligible for free or reduced-price lunch during the experiment.

4 Own motivation and educational success

4.1 Sample selection and summary statistics

In this Section, I study the importance of children's academic motivation for their own educational success. For this purpose, I select the sample that maximizes the number of children observed with this personality measure. Of the total number of 11,601 students in Project STAR, 9,932 participated in the experiment at some point during grades 1-3, with the others dropping out after kindergarten. 9,072 of these participants are observed with a motivation score in at least one of these grades and are included in what I will refer to as the own motivation sample. The missing information for the

other, excluded 860 students is mostly due to data processing issues (see Word et al., 1990).

In my analysis, I am interested in the predictive power of the relatively stable part of children's motivation. I therefore construct my main independent variable as the average motivation of each student during grades 1-3. Specifically, I first standardize the motivation scores for each grade to have mean 0 and SD 1. I then average the available scores for each student across grades and standardize the resulting composite again. This lets me interpret regression coefficients as the predicted change in the outcome if academic motivation in early elementary school increases by 1 SD. Averaging across grades in this way is also beneficial because it reduces measurement error.

The regressions below relate children's academic motivation to four sets of outcomes. First, I consider achievement in early elementary school as measured by average test scores across grades 1-3 in reading and math. Second, I study achievement in middle school as measured by average test scores across grades 5-8 in reading and math. Third, I examine children's longer-term educational success as captured by their high school GPA, high school graduation, and taking of an ACT or SAT test. Finally, I study classroom behavior as captured by the eight indices constructed from teacher ratings in fourth and eighth grade. To facilitate interpretation, I standardize all measures of achievement and classroom behavior to have mean 0 and SD 1.

Table 1 shows summary statistics for the own motivation sample. Due to the fact that Project STAR oversampled schools in poor neighborhoods, students are disproportionately likely to be black and eligible for free or reduced-price lunch. 82 percent of students graduated from high school and 38 percent took an ACT or SAT test around the age of 18. Note that not all students are observed with all outcomes due to different data collection procedures and sample attrition, see Online Appendix A for details.

4.2 Regression specification

To test whether academic motivation in early elementary school predicts educational success, I estimate ordinary least squares regressions of the following form:

$$y_{is} = \alpha + \beta \text{MOTIV}_i^{G1-G3} + X_i \gamma + \lambda_s + \varepsilon_{is}, \qquad (1)$$

where *i* denotes students and *s* denotes school-by-entry-grade cells, that is, the Project STAR randomization blocks. y_{is} is the outcome of interest. MOTIV_i^{G1-G3} is student *i*'s average academic motivation across grades 1-3. X_i is a vector of socio-demographic

characteristics shown in Table 1.⁹ λ_i is a vector of school-by-entry-grade dummies, which account for differences between students entering the various schools participating in Project STAR in different grades. Finally, ε_{is} is the error term. In all regressions, I cluster standard errors at the level of school-by-entry-grade.

4.3 Main results

Panel A of Table 2 reports estimates based on Equation 1 and reveals that motivation in early elementary school is predictive of both contemporaneous and later educational success. In particular, a 1 SD higher motivation score is associated with 0.05 SD (0.06 SD) higher reading (math) scores in early elementary and middle school and a 0.3 points increased high school GPA, which is measured on a scale from 0-100. Strikingly, motivation in early elementary school also predicts college intent more than ten years later, as students with a 1 SD higher motivation score are 1.7 percentage points (4.4 percent) more likely to take an ACT or SAT test around age 18.

An important question is whether the estimates shown in Panel A reflect impacts of motivation or whether they are due to underlying academic ability, which likely correlates with both motivation and educational success. In Panel B, I address this question by adding controls for reading and math achievement in kindergarten to the regressions as proxies for academic ability. Interestingly, this leaves the results largely unchanged compared to Panel A. This suggests that the estimates in Table 2 indeed capture effects of motivation and not just impacts of correlated academic ability.¹⁰

Table 3 presents results for teacher-rated classroom behavior and reveals that motivation is predictive of all observed measures of good behavior. For example, Panel A shows that a 1 SD higher motivation score is associated with 0.11 SD higher effort and 0.09 SD higher discipline in fourth grade. The associations are slightly weaker for the eighth-grade measures, which could reflect either fade-out or the fact that the questions on which teachers rated students were slightly different in that grade. Panel B reveals that the estimates for both grades are robust to controlling for achievement in kinder-

⁹ As can be seen in Table 1, there are some missing values in these control variables. In order not to reduce sample size unnecessarily, in all regressions in this paper I impute missing values in controls at the sample mean and include separate dummies for missing values on each control variable. Results are virtually identical if I instead exclude students with missing information on socio-demographic characteristics from the sample.

¹⁰I make this argument more formally in Online Appendix Table B.1, where I use the method proposed by Oster (2019) to assess how large omitted variable bias due to factors such as academic ability would have to be in order to drive the coefficients in Table 2 to zero. I find that selection on unobservables would have to be several times larger than selection on achievement in kindergarten in order to explain away the coefficient on motivation in the regressions.

garten. Taken together, the results in this table confirm that the SCAMIN motivation score captures a dimension of personality that is reflected in actual behaviors. This finding is interesting in its own right but also because previous studies have shown that these behaviors predict later educational attainment and earnings among participants in Project STAR (Chetty et al., 2011; Bietenbeck, 2020).

4.4 Further results

I now summarize results from some additional analyses. First, Online Appendix Table B.2 shows estimates of the relationship between motivation and educational success separately for different groups of students. The results reveal that for most outcomes, the predicted improvement due to a 1 SD higher motivation is larger for girls than for boys, larger for black students than for non-black students, and larger for free-lunch students than for non-free-lunch students. Second, Online Appendix Table B.3 presents results from sensitivity checks, which reveal that the associations between motivation and educational success presented in Table 2 are robust to restricting the sample to students observed with achievement in kindergarten¹¹ and to controlling for students' academic self-concept in grades 1-3 as measured by the SCAMIN.

5 Peer motivation and educational success

5.1 Sample selection and summary statistics

In this Section, I study how exposure to motivated peers affects children's educational success. Specifically, I estimate causal spillover effects on students who first entered Project STAR in second or third grade. The new classmates of these entrants had participated in the experiment and written the SCAMIN test in the previous (first or second) grade, which allows me to observe their academic motivation. As students in Project STAR were randomly assigned to classes within school upon entry, this means that there is random and observable variation in the motivation of second- and third-grade entrants' classmates, which I use to estimate spillover effects.

A total of 2,962 students entered Project STAR in second or third grade. For 2,868 of these students, I observe the motivation of at least some of their new classmates, and these entrants constitute what I will refer to as the peer motivation sample. I

¹¹Measures of kindergarten achievement are available only for those students who participated in Project STAR in kindergarten. As described in the notes to Table 2, in the main regressions I impute missing values in kindergarten achievement for the other students who joined the experiment after kindergarten at the sample mean.

construct peer motivation as the average motivation of entrants' classmates measured at the end the previous school year, thus ensuring that peer motivation is predetermined relative to the assignment of entrants to classes. In a similar fashion, I also construct averages of classmates' socio-demographic characteristics and their reading and math achievement in the previous grade, which I use as controls in some regressions. To facilitate interpretation of results, I standardize peer motivation and peer achievement to have mean 0 and SD 1.

In line with the bulk of the previous research on peer effects, the main specifications focus on spillover effects on contemporaneous outcomes. Specifically, I estimate how exposure to motivated peers affects entrants' reading and math achievement at the end of their first year in Project STAR. In additional analyses, I also examine impacts on longer-term outcomes, including reading and math achievement in middle school, high school graduation and GPA, and college-test taking. For ease of interpretation, I standardize all achievement outcomes to have mean 0 and SD 1.

Table 4 shows summary statistics for the peer motivation sample. Compared to the larger own motivation sample used in Section 4, the students in this sample are even more disadvantaged: for example, they are more likely to be eligible for free or reduced-price lunch (66 versus 60 percent) and less likely to graduate from high school (73 vs 82 percent) and to take an ACT or SAT test (26 versus 38 percent). Note that like with the own motivation sample, not all students are observed with all outcomes due to limited data collection or attrition from the sample. Later on, I show in a robustness check that this missing data problem is not driving my results.

5.2 Regression specification

I estimate regressions of the following form:

$$y_{ics} = \delta \overline{\text{MOTIV}}_c^{G-1} + \phi \text{SMALL}_c + X_i \eta + \overline{Z}_c \rho + \omega_s + \mu_{ics}, \qquad (2)$$

where *i* denotes students, *c* denotes classes, and *s* denotes school-by-entry grade cells. y_{ics} is the outcome of interest. $\overline{\text{MOTIV}}_c^{G-1}$ is the average motivation of students in class *c* who participated in Project STAR in the previous grade (G-1). SMALL_c is a dummy for assignment to a small class, the original treatment of interest in Project STAR. X_i is a vector of student socio-demographic characteristics and \overline{Z}_c is a vector of peer characteristics shown in Table 4. Finally, ω_s is a vector of school-by-entry-grade dummies that accounts for fixed differences between randomization pools and μ_{ics} is the error term. For all regressions, I compute standard errors that allow for clustering at the level of school-by-entry-grade.

Equation 2 corresponds to a linear-in-means model, which is the most widely estimated model of peer effects (Sacerdote, 2011). The main coefficient of interest, δ , captures the causal impact of exposure to motivated peers under the assumption that variation in peer motivation is random within school-by-entry-grade cells, an assumption that I support with empirical evidence below. Since peer motivation is correlated with other peer characteristics, an obvious question is whether δ captures spillovers from motivation or from such other characteristics. I address this question by controlling for peer achievement and peer socio-demographic characteristics in some of my regressions. If the estimates are robust to the inclusion of these controls, this suggests that δ indeed captures spillovers from peer motivation, rather than from correlated observed and unobserved factors (Altonji, Elder, and Taber, 2005; Oster, 2019).

5.3 Evidence on random assignment

Previous studies using data from Project STAR provide detailed evidence that students were randomly assigned to classes within school upon entry, see especially Chetty et al. (2011) and Sojourner (2013). Here, I complement this evidence by showing that peer motivation is unrelated to predetermined characteristics of students entering the experiment in second or third grade.

Table 5 reports results from regressions like in Equation 2 in which the dependent variables are students' predetermined socio-demographic characteristics (columns 1-4). As a further dependent variable, I constructed a measure of predicted achievement that combines these socio-demographic characteristics such that they optimally predict students' reading and math scores (column 5).¹² Panel A shows estimates from separate regressions for peer motivation and, to further buttress the results, peer achievement in reading and math. Panel B shows estimates from specifications in which these three peer variables enter simultaneously instead. Across all regressions, most of the coefficients on the peer variables are close to zero and not statistically significant at conventional levels. In the regressions in Panel B, the coefficients are also jointly insignificant. This strongly suggests that second- and third-grade entrants in Project STAR were indeed randomized to classes within school upon entry.

In Online Appendix B, I present two further pieces of evidence in favor of random assignment. First, following Chetty et al. (2011), Online Appendix Table B.4 shows

¹²Specifically, I predict achievement from a regression of the averaged reading and math score at the end of students' first year in Project STAR on the four socio-demographic characteristics and school-by-entry-grade fixed effects.

that class dummies do not jointly predict predetermined characteristics of entrants, as should be the case if they were randomized into classes. Second, following Feld and Zölitz (2017), I ran separate regressions of these characteristics on class dummies for each school-by-entry-grade cell. After each regression, I conducted an F test for the joint significance of the class dummies and collected the p-value. Under random assignment, these p-values should be distributed roughly uniformly, and Online Appendix Figure B.1 shows that this is indeed the case. Moreover, the shares of p-values below certain confidence levels should be close to this level (for example, about five percent of p-values should be below 0.05), and Online Appendix Table B.5 confirms this. This evidence provides strong additional support for the assumption that second- and third-grade entrants were randomly assigned to classes within school in Project STAR.

5.4 Effects on contemporaneous achievement

Table 6 reports estimates of the effect of exposure to motivated peers on achievement in reading and math at the end of entrants' first year in Project STAR. Column 1 shows that having classmates with a 1 SD higher average motivation raises own reading scores by 0.077 SD. Column 4 shows an effect on math scores that is also positive but smaller at 0.034 SD and not statistically significant at conventional levels.

Columns 2 and 5 add controls for peer achievement to these specifications. If spillovers from motivated peers were mainly due to correlated peer ability, we would expect to see a reduction in the size of the coefficient on peer motivation in these regressions. However, the estimates are largely unchanged, suggesting that this is not the case. Columns 3 and 6 show that the results are also robust to controlling for classmates' socio-demographic characteristics, spillovers from which have been studied extensively in the previous literature (e.g. Hoxby, 2000; Whitmore, 2005; Lavy and Schlosser, 2011). In this most demanding specification, a 1 SD increase in peer motivation is estimated to raise own reading scores by 0.071 SD. For comparison, the estimated effect of a 1 SD increase in peer reading achievement is 0.147 SD in the same regression.

The fact that the estimates in Table 6 change only very little when controls for other peer variables are added to the regressions suggests that they capture a true personality spillover from classmates' motivation, rather than a spillover from correlated unobserved factors. I provide formal evidence in support of this argument in Online Appendix Table B.6, where I use the method developed by Oster (2019) to assess how large omitted variable bias would have to be in order to drive the estimated effect of peer motivation on reading scores to zero. I find that under standard assumptions, selection on unobservables would have to be more than twice as large as selection on observed peer achievement and socio-demographic characteristics to explain away the effect. As that is relatively unlikely (Oster, 2019), this finding supports the interpretation of my estimates as capturing spillovers from peer motivation.

In additional analyses, I examine potential heterogeneities and non-linearities in the effect of peer motivation on achievement. I briefly summarize the results in what follows. First, Figure 1 visualizes the estimates from Table 6 and reveals that the effect of peer motivation on reading scores is roughly linear. Second, Online Appendix Table B.7 shows that the effect is larger for boys than for girls, but that it does not differ much by race or by eligibility for free lunch. Third, Online Appendix Table B.8 reports estimates from specifications in which peer motivation is interacted with the small-class dummy and specifications in which the sample is restricted to students in regular-sized classes. The regression estimates point toward a larger effect of peer motivation on achievement in regular-sized classes, although differences by class size are never statistically significant at conventional levels.

Finally, in Online Appendix Table B.9 I move beyond the linear-in-means model of peer effects and investigate how exposure to peers with very high motivation ("shining lights") and exposure to peers with very low motivation ("bad apples") affects achievement. To that end, I replace the average peer motivation in Equation 2 with the shares of classmates with top-tercile and bottom-tercile motivation scores. The main take-away from these estimates is that having a high share of classmates with bottom-tercile motivation significantly reduces reading achievement, whereas the impact of classmates with top-tercile motivation is positive but smaller in absolute value and not statistically significant at conventional levels.

5.5 Effects on own motivation and self-concept

An intriguing idea is that peer motivation might influence children's own personality. In Table 7, I explore such spillovers by estimating the effect of peer motivation on entrants' own motivation and self-concept at the end of their first year in Project STAR. To the best of my knowledge, these are the first estimates of spillovers from peer personality on own personality in the literature. The estimated effect of peer motivation in both regressions is very close to zero, showing that peer motivation does not affect own motivation or self-concept, at least as measured by the SCAMIN.

5.6 Effects on longer-term educational success

Given that peer motivation raises contemporaneous achievement, an obvious question is whether it also affects students' longer-term educational success. Table 8 shows that this appears not to be the case: there is little indication that peer motivation affects middle school test scores, high school outcomes, or college-test taking. When interpreting these estimates, it is important to realize that they likely capture the effects of a relatively short exposure to more motivated peers during elementary school. Specifically, when Project STAR ended after third grade, students were redistributed to ordinary classes. While I do not observe class composition beyond third grade, this re-shuffling probably meant that later peer motivation was at most weakly related to peer motivation in second or third grade. In turn, this implies that there is no mechanical longer-term impact of peer motivation in early elementary school due to classmates staying together, and consequently the estimates in Table 8 capture the effects of differential exposure to more motivated peers for only one or two years.

5.7 Mechanisms

The results above show that peer motivation raises contemporaneous reading achievement, but that it does not affect longer-term educational success. Peer motivation also does not change own motivation and self-concept. In what follows, I briefly discuss the potential mechanisms underlying these findings.

First, it is important to note that the pattern of impacts is consistent with previous studies on childhood interventions, which have found that treatments are particularly successful at changing longer-term outcomes if they affect children's personality (e.g. Heckman, Pinto, and Savelyev, 2013), and with earlier papers on peer effects, which have argued that school peers influence children's long-term educational and labor market success mainly via their impact on non-cognitive skills (e.g. Carrell, Hoekstra, and Kuka, 2018; Bietenbeck, 2020). This suggests that the absence of longer-term impacts of peer motivation in the analysis above is due to the lack of an effect on own personality. Put differently, it appears that the contemporaneous impact on reading scores by itself is simply not large enough to generate measurable long-term effects.

Second, given that peer motivation does not affect own motivation and self-concept, what explains the rise in reading scores? Perhaps the most likely mechanism is that having more motivated peers leads to an improved learning environment in the classroom. As shown in Section 4, motivated students are more disciplined and generally show better behavior in the classroom according to their teachers. This implies that entrants whose peers are more motivated likely experience less distraction from them, which in turn could account for the documented increase in reading scores.

5.8 Robustness

I now summarize the results from robustness checks that address several potential concerns about my results. First, I study effects on many different outcomes, which raises the possibility that the only statistically significant effect on contemporaneous reading achievement represents a chance finding. To mitigate this threat, Online Appendix Table B.10 reports estimates of the effect of peer motivation on word study skills, which are closely related to reading skills and which were also assessed by the Stanford Achievement Test.¹³ The results show that a 1 SD increase in peer motivation raises word study skills scores by a highly statistically significant 0.081 SD, an effect that is almost identical in size to the impact on reading scores. I moreover confirmed that the effects of peer motivation on reading scores and word study skills scores remain statistically significant when I correct for multiple hypothesis testing using the method developed by Romano and Wolf (2005a,b), see Online Appendix Table B.11.

Second, not all outcomes are observed for all students in the sample, which opens up the possibility that my results are biased by selective attrition. To address this threat, Online Appendix Table B.12 shows estimates of the effect of peer motivation on indicators for being observed with each of the outcomes studied in Tables 6 and 8. The coefficients from the regressions of contemporaneous achievement and most longer-term outcomes are close to zero and not statistically significant at conventional levels, showing that the likelihood of being observed with these outcomes does not systematically vary with peer motivation.¹⁴ This finding strongly suggests that selective attrition is not driving my results.

Third, academic motivation is usually not observed for all classmates of entrants, partly due to data processing issues (see Word et al., 1990). This introduces measurement error, which could bias my estimates. To mitigate this concern, Online Appendix Table B.13 shows results from regressions in which the sample is restricted to entrants

¹³The correlation coefficient between reading scores and word study skills scores is 0.88. For completeness, Online Appendix Table B.10 also shows the effect on listening skills, the fourth and final skills domain assessed by the Stanford Achievement Test in both second and third grade (the correlation coefficient between reading scores and listening scores is 0.64). I do not include word study skills and listening skills in the main analysis for conciseness and in order to keep in line with the previous literature on Project STAR, which has focused almost exclusively on reading and math.

¹⁴There is a marginally statistically significant negative effect on being observed with middle school test scores. This could potentially explain the negative (but insignificant) point estimates of peer motivation on middle school reading and math achievement in Table 8.

for whom information on personality is available for most classmates. The effect of peer motivation on reading scores in these regressions is very similar to the one reported in Table 6, although the estimate is less precise due to the lower number of observations. Finally, Online Appendix Table B.14 shows that results are robust to controlling for peer self-concept as measured by the SCAMIN.

6 Conclusion

A growing literature in economics has studied the importance of personality for educational success, but has mostly focused on preference parameters and personality traits. Moreover, despite extensive evidence that the social environment matters for performance in school, only very few studies have examined spillovers from classmates' personality. In this paper, I contribute to this research by showing that motivation, an important facet of personality, matters for own and peers' educational success.

In the first part of the paper, I show that academic motivation in early elementary school, as measured by a standardized psychological test, is predictive of contemporaneous and future achievement in reading and math, high school GPA, and college-test taking around age 18. These associations hold even when I control for achievement in kindergarten, which suggests that they are not due to correlated academic ability. In the second part of the paper, I show that exposure to motivated peers causally affects reading achievement in elementary school. This peer effect operates over and above any spillovers from classmates' past achievement and socio-demographic composition, which suggests that it reflects a true personality spillover. I also show that peer motivation does not affect own motivation, a finding that likely explains the lack of any longer-term impacts of short-term exposure to motivated peers.

Previous research in economics has shown that personality is malleable especially during childhood (Kautz et al., 2014) and that targeted programs can effectively change facets of personality such as patience (Alan and Ertac, 2018), grit (Alan, Boneva, and Ertac, 2019), prosociality (Kosse et al., 2020), and socio-emotional skills (Sorrenti et al., 2020). Moreover, studies in psychology have documented that interventions such as helping students set goals or instructing teachers to relate lesson content to students' experiences can improve their motivation and achievement (Lazowski and Hulleman, 2016). An important implication of my findings is that the benefits of such interventions extend beyond the targeted students, as the generated improvements in these facets of personality will spill over onto their peers.

References

- Alan, S., T. Boneva, and S. Ertac. 2019. "Ever Failed, Try Again, Succeed Better: Results from a Randomized Educational Intervention on Grit." *The Quarterly Journal of Economics* 134:1121–1162.
- Alan, S., and S. Ertac. 2018. "Fostering Patience in the Classroom: Results from Randomized Educational Intervention." Journal of Political Economy 126:1865–1911.
- Almlund, M., A.L. Duckworth, J. Heckman, and T. Kautz. 2011. "Personality Psychology and Economics." In E. A. Hanushek, S. Machin, and L. Woessmann, eds. *Handbook of the Economics of Education*. Elsevier, vol. 4, pp. 1–181.
- Altonji, J.G., T.E. Elder, and C.R. Taber. 2005. "Selection on Observed and Unobserved Variables: Assessing the Effectiveness of Catholic Schools." *Journal of Political Economy* 113:151–184.
- Ballis, B. 2020. "Does Peer Motivation Impact Schooling Investments? Evidence from DACA." Working paper.
- Bietenbeck, J. 2020. "The Long-Term Impacts of Low-Achieving Childhood Peers: Evidence from Project STAR." Journal of the European Economic Association 18:392–426.
- Booij, A.S., E. Leuven, and H. Oosterbeek. 2017. "Ability Peer Effects in University: Evidence from a Randomized Experiment." *The Review of Economic Studies* 84:547–578.
- Borghans, L., A.L. Duckworth, J.J. Heckman, and B. Ter Weel. 2008. "The Economics and Psychology of Personality Traits." *Journal of Human Resources* 43:972–1059.
- Brenoe, A.A., and U. Zölitz. 2019. "Exposure to More Female Peers Widens the Gender Gap in STEM Participation." *Journal of Labor Economics* 38:1009–1054.
- Cadena, B.C., and B.J. Keys. 2015. "Human Capital and the Lifetime Costs of Impatience." American Economic Journal: Economic Policy 7:126–153.
- Carrell, S.E., M. Hoekstra, and E. Kuka. 2018. "The Long-Run Effects of Disruptive Peers." American Economic Review 108:3377–3415.
- Carrell, S.E., and M.L. Hoekstra. 2010. "Externalities in the Classroom: How Children Exposed to Domestic Violence Affect Everyone's Kids." American Economic Journal: Applied Economics 2:211–228.
- Chetty, R., J.N. Friedman, N. Hilger, E. Saez, D.W. Schanzenbach, and D. Yagan. 2011. "How Does Your Kindergarten Classroom Affect Your Earnings? Evidence from Project STAR." *The Quarterly Journal of Economics* 126:1593–1660.
- Clarke, D., J.P. Romano, and M. Wolf. 2019. "The Romano-Wolf Multiple Hypothesis Correction in Stata." *Stata Journal* forthcoming.
- Davis, T.M., and J.M. Johnston. 1987. "On the Stability and Internal Consistency of the Self-Concept and Motivation Inventory: Preschool/Kindergarten Form." *Psychological Reports* 61:871–874.

- Davis, T.M., P.A. Sellers, and J.M. Johnston. 1988. "The Factor Structure and Internal Consistency of the Self-Concept and Motivation Inventory: What Face Would You Wear? Preschool/Kindergarten Form." *Educational and Psychological Measurement* 48:237–246.
- Drummond, R.J., and W.G. McIntire. 1975. "Note on Test-Retest Reliability of the Self-Concept and Motivation Inventory." *Psychological Reports* 36:563–566.
- Duckworth, A.L., C. Peterson, M.D. Matthews, and D.R. Kelly. 2007. "Grit: Perseverance and Passion for Long-Term Goals." Journal of Personality and Social Psychology 92:1087–1101.
- Dunifon, R., and G.J. Duncan. 1998. "Long-run effects of motivation on labor-market success." Social Psychology Quarterly, pp. 33–48.
- Feld, J., and U. Zölitz. 2017. "Understanding Peer Effects: On the Nature, Estimation, and Channels of Peer Effects." *Journal of Labor Economics* 35:387–428.
- Figlio, D.N. 2007. "Boys Named Sue: Disruptive Children and Their Peers." Education Finance and Policy 2:376–394.
- Finn, J.D., J. Boyd-Zaharias, R.M. Fish, and S.B. Gerber. 2007. "Project STAR and Beyond: Database User's Guide." Report, HEROS Incorporated.
- Finn, J.D., and D. Cox. 1992. "Participation and Withdrawal among Fourth-Grade Pupils." American Educational Research Journal 29:141–162.
- Gensowski, M. 2018. "Personality, IQ, and Lifetime Earnings." Labour Economics 51:170–183.
- Golsteyn, B., A. Non, and U. Zölitz. 2020. "The Impact of Peer Personality on Academic Achievement." *Journal of Political Economy* forthcoming.
- Golsteyn, B.H., H. Grönqvist, and L. Lindahl. 2014. "Adolescent Time Preferences Predict Lifetime Outcomes." The Economic Journal 124:F739–F761.
- Heckman, J., R. Pinto, and P. Savelyev. 2013. "Understanding the Mechanisms through which an Influential Early Childhood Program Boosted Adult Outcomes." *American Economic Review* 103:2052–2086.
- Heckman, J.J., T. Jagelka, and T.D. Kautz. 2019. "Some Contributions of Economics to the Study of Personality." NBER Working Paper No. 26459.
- Hoxby, C. 2000. "Peer Effects in the Classroom: Learning from Gender and Race Variation." NBER Working Paper No. 7867.
- Kautz, T., J.J. Heckman, R. Diris, B.t. Weel, and L. Borghans. 2014. "Fostering and Measuring Skills: Improving Cognitive and Non-Cognitive Skills to Promote Lifetime Success." OECD Education Working Paper No. 110.
- Koch, A., J. Nafziger, and H.S. Nielsen. 2015. "Behavioral economics of education." Journal of Economic Behavior & Organization 115:3–17.
- Komarraju, M., and S.J. Karau. 2005. "The Relationship between the Big Five Personality Traits and Academic Motivation." *Personality and Individual Differences* 39:557–567.

- Kosse, F., T. Deckers, P. Pinger, H. Schildberg-Hörisch, and A. Falk. 2020. "The Formation of Prosociality: Causal Evidence on the Role of Social Environment." *Journal of Political Economy* 128:434–467.
- Krueger, A.B. 1999. "Experimental Estimates of Education Production Functions." The Quarterly Journal of Economics 114:497–532.
- Krueger, A.B., and D.M. Whitmore. 2001. "The Effect of Attending a Small Class in the Early Grades on College-Test Taking and Middle School Test Results: Evidence from Project STAR." The Economic Journal 111:1–28.
- Lavy, V., and A. Schlosser. 2011. "Mechanisms and Impacts of Gender Peer Effects at School." American Economic Journal: Applied Economics 3:1–33.
- Lavy, V., O. Silva, and F. Weinhardt. 2012. "The Good, the Bad, and the Average: Evidence on Ability Peer Effects in Schools." *Journal of Labor Economics* 30:367–414.
- Lazowski, R.A., and C.S. Hulleman. 2016. "Motivation Interventions in Education: A Meta-Analytic Review." *Review of Educational Research* 86:602–640.
- Mas, A., and E. Moretti. 2009. "Peers at Work." American Economic Review 99:112–145.
- McIntire, W.G., and R.J. Drummond. 1976. "The Structure of Self-Concept in Second and Fourth Grade Children." *Educational and Psychological Measurement* 36:529–536.
- Measelle, J.R., O.P. John, J.C. Ablow, P.A. Cowan, and C.P. Cowan. 2005. "Can Children Provide Coherent, Stable, and Valid Self-Reports on the Big Five Dimensions? A Longitudinal Study From Ages 5 to 7." Journal of Personality and Social Psychology 89:90–106.
- Milchus, N.J., G.A. Farrah, and W. Reitz. 1968. *The Self-concept and Motivation Inventory: What Face Would You Wear?*. Dearborn Heights, MI: Person-O-Metrics.
- Murphy, P.K., and P.A. Alexander. 2000. "A Motivated Exploration of Motivation Terminology." Contemporary Educational Psychology 25:3–53.
- Oster, E. 2019. "Unobservable Selection and Coefficient Stability: Theory and Evidence." Journal of Business & Economic Statistics 37:187–204.
- Paloyo, A.R. 2020. "Peer Effects in Education: Recent Empirical Evidence." In S. Bradley and C. Green, eds. *The Economics of Education (Second Edition)*. Academic Press, pp. 291–305.
- Piatek, R., and P. Pinger. 2016. "Maintaining (Locus of) Control? Data Combination for the Identification and Inference of Factor Structure Models." *Journal of Applied Econometrics* 31:734–755.
- Poropat, A.E. 2009. "A Meta-Analysis of the Five-Factor Model of Personality and Academic Performance." Psychological Bulletin 135:322–338.
- Roberts, B.W. 2006. "Personality Development and Organizational Behavior." Research in Organizational Behavior 27:1–40.

- Roberts, B.W., and W.F. DelVecchio. 2000. "The rank-order consistency of personality traits from childhood to old age: a quantitative review of longitudinal studies." *Psychological bulletin* 126:3–25.
- Roberts, B.W., P. Harms, J.L. Smith, D. Wood, and M. Webb. 2006. "Using Multiple Methods in Personality Psychology." In M. Eid and E. Diener, eds. *Handbook of Multimethod Measurement in Psychology*. American Psychological Association, pp. 321–335.
- Romano, J.P., and M. Wolf. 2005a. "Exact and Approximate Stepdown Methods for Multiple Hypothesis Testing." *Journal of the American Statistical Association* 100:94–108.
- 2005b. "Stepwise Multiple Testing as Formalized Data Snooping." *Econometrica* 73:1237–1282.
- Sacerdote, B. 2011. "Peer Effects in Education: How Might They Work, How Big Are They and How Much Do We Know Thus Far?" In E. A. Hanushek, S. Machin, and L. Woessmann, eds. Handbook of the Economics of Education. Elsevier, vol. 3, pp. 249–277.
- Schanzenbach, D.W. 2006. "What Have Researchers Learned from Project STAR?" Brookings Papers on Education Policy 9:205–228.
- Segal, C. 2012. "Working when no one is watching: Motivation, test scores, and economic success." *Management Science* 58:1438–1457.
- Shavelson, R.J., J.J. Hubner, and G.C. Stanton. 1976. "Self-Concept: Validation of Construct Interpretations." *Review of Educational Research* 46:407–441.
- Shure, N. 2017. "Non-Cognitive Peer Effects in Secondary Education." Unpublished.
- Sojourner, A. 2013. "Identification of Peer Effects with Missing Peer Data: Evidence from Project STAR." *The Economic Journal* 123:574–605.
- Sorrenti, G., U. Zölitz, D. Ribeaud, and M. Eisner. 2020. "The Causal Impact of Socio-Emotional Skills Training on Educational Success." CEPR Discussion Paper No. 14523.
- Steinmayr, R., and B. Spinath. 2009. "The Importance of Motivation as a Predictor of School Achievement." *Learning and Individual Differences* 19:80–90.
- Whitmore, D. 2005. "Resource and Peer Impacts on Girls' Academic Achievement: Evidence from a Randomized Experiment." *American Economic Review* 95:199–204.
- Winter, D.G., O.P. John, A.J. Stewart, E.C. Klohnen, and L.E. Duncan. 1998. "Traits and motives: Toward an integration of two traditions in personality research." *Psychological Review* 105:230–250.
- Wong, M.M., and M. Csikszentmihalyi. 1991. "Motivation and Academic Achievement: The Effects of Personality Traits and the Duality of Experience." *Journal of Personality* 59:539– 574.
- Word, E., J. Johnston, H.P. Bain, D.B. Fulton, C.M. Achilles, M.N. Lintz, J. Folger, and C. Breda. 1990. "The State of Tennessee's Student/Teacher Achievement Ratio (STAR) Project: Technical Report 1985-1990." Report, Tennessee State Department of Education.

Figures and Tables



Figure 1: Peer motivation and entry-grade achievement

Notes: The figure shows estimates of the effect of peer motivation on achievement in reading and math at the end of entrants' first year in Project STAR. To construct these plots, I first residualize achievement scores and peer motivation on the controls included in the specifications in columns 3 and 6 of Table 6. I then group residualized peer motivation into ten equal-sized bins and plot the mean of the residualized achievement scores for each bin. The regression line in each plot is based on the underlying individual-level data and thus visualizes the corresponding regression in Table 6.

	Mean	SD	Ν
Socio-demographic characteris	tics		
Male	0.53	0.50	9,072
Black	0.35	0.48	9,054
Free lunch	0.60	0.49	$8,\!978$
Age in 1985	5.75	0.57	9,065
Achievement in kindergarten			
Reading score	0.00	1.00	4,174
Math score	0.00	1.00	4,218
Academic motivation			
Motivation in grades 1-3	0.00	1.00	9,072
Educational outcomes			
Reading scores in grades 1-3	0.00	1.00	8,530
Math scores in grades 1-3	0.00	1.00	8,678
Reading scores in grades 5-8	0.00	1.00	7,497
Math scores in grades 5-8	0.00	1.00	7,493
High school GPA $(0-100)$	83.50	7.57	3,360
High school graduation	0.82	0.39	4,368
Took ACT/SAT	0.38	0.48	9,072
Classroom behavior			
Effort in grade 4	0.00	1.00	2,212
Initiative in grade 4	0.00	1.00	2,212
Discipline in grade 4	0.00	1.00	2,212
Value in grade 4	0.00	1.00	2,212
Effort in grade 8	0.00	1.00	$2,\!693$
Initiative in grade 8	0.00	1.00	$2,\!693$
Discipline in grade 8	0.00	1.00	$2,\!693$
Value in grade 8	0.00	1.00	$2,\!693$

Table 1: Summary statistics for the own motivation sample

Notes: The table shows means and standard deviations and the number of students observed with each variable for the 9,072 students included in the own motivation sample.

	1						
	Grae	les $1-3$	Grae	les $5-8$	High :	school	College
	reading	math scores	reading	math scores	GPA	gradua- tion	took ACT /SAT
	(1)	(2)	(3)	(4)	(5)	(9)	(1)
Panel A: baseline specifica Motivation in grades 1-3	$tion \\ 0.050^{***}$	0.056^{***}	0.054^{***}	0.056^{***}	0.292^{*}	0.008	0.017^{***}
)	(0.011)	(0.012)	(0.014)	(0.013)	(0.149)	(0.007)	(0.005)
Observations	(8,530)	8,678	7,497	7,493	3,360	4,368	9,072
Panel B: specification with	<i>i</i> controls for	achievement in	kindergarten				
Motivation in grades 1-3	0.043^{***}	0.049^{***}	0.049^{***}	0.051^{***}	0.269^{**}	0.008	0.015^{***}
	(0.010)	(0.010)	(0.012)	(0.012)	(0.133)	(0.007)	(0.004)
Observations	8,530	8,678	7,497	7,493	3,360	4,368	9,072
Notes: The table shows estin motivation, averaged across g	mates from re grades 1-3. A	gressions of the Il regressions in F	outcome vari. Panels A and	ables indicated in B control for sch	ool-by-entry-g	aders on stue ade fixed effec	lents' academic ts, dummies for
achievement in kindergarten.	For students	with missing info	and age. rues rmation on ki	ndergarten achiev	rement, test sco	ores are impute	d at the sample
mean, with regressions contr	colling for two	separate dumm	iies indicating	imputation of k	indergarten re	ading and ma	th achievement.
Standard errors in parenthese	ss are clustered	d by school-by-en	try-grade. $* p$	<0.10, ** p<0.05	, *** p < 0.01.		

Table 2: Own motivation and educational success

		Gra	de 4			Gra	ıde 8	
	effort (1)	initiative (2)	discipline (3)	value (4)	effort (5)	initiative (6)	discipline (7)	value (8)
Panel A: baseline specifica Motivation in grades 1-3	<i>ution</i> 0.105***	0.084^{***}	0.090***	0.112^{***}	0.062^{***}	0.039^{*}	0.066^{***}	0.079***
I	(0.027)	(0.026)	(0.028)	(0.031)	(0.022)	(0.023)	(0.024)	(0.024)
Observations	2,212	2,212	2,212	2,212	2,693	2,693	2,693	2,693
Panel B: specification with	h controls fo	r achievemen	t in kinderga	rten				
Motivation in grades 1-3	0.113^{***}	0.093^{***}	0.094^{***}	0.116^{***}	0.068^{***}	0.045^{**}	0.068^{***}	0.081^{***}
	(0.026)	(0.025)	(0.029)	(0.030)	(0.021)	(0.021)	(0.024)	(0.024)
Observations	2,212	2,212	2,212	2,212	2,693	2,693	2,693	2,693
<i>Notes:</i> The table shows estimution motivation, averaged across grade black and elicibility for	mates from grades 1-3. 2	regressions of All regressions htred-bride lut	the outcome in Panels A whand age	variables inc and B contro Regressions	licated in the ol for school-b in Panel R ac	column head y-entry-grade	ders on studer e fixed effects, ntrol for readi	its' academic dummies for no and math
achievement in kindergarten.	For students	s with missing	information c	in kindergart	en achievemei	nt, test scores	are imputed a	at the sample
mean, with regressions conti	rolling for tv	vo separate dı	ummies indica	ating imputa	tion of kinde	rgarten readi	ng and math	achievement.
Standard errors in parenthese	es are cluster	ed by school-b	by-entry-grade	. * <i>p</i> <0.10, *	* p<0.05, ***	p < 0.01.		

Table 3: Own motivation and classroom behavior

	Mean	SD	Ν	
Socio-demographic characteris	tics			
Male	0.55	0.50	2,861	
Black	0.42	0.49	2,766	
Free lunch	0.66	0.47	2,730	
Age in 1985	6.01	0.70	$2,\!845$	
Peer motivation and other pee	r character	ristics		
Peer motivation	0.00	1.00	2,868	
Peer reading achievement	0.00	1.00	2,841	
Peer math achievement	0.00	1.00	2,850	
Peer share male	0.51	0.11	2,868	
Peer share black	0.42	0.43	2,868	
Peer share free lunch	0.61	0.30	2,868	
Entry-grade achievement				
Reading score	0.00	1.00	2,185	
Math score	0.00	1.00	2,196	
Entry-grade own personality				
Own motivation	0.00	1.00	2,276	
Own self-concept	0.00	1.00	2,276	
Longer-term educational outcomes				
Reading scores in grades 5-8	0.00	1.00	2,118	
Math scores in grades 5-8	0.00	1.00	2,119	
High school GPA (0-100)	81.50	7.46	665	
High school graduation	0.73	0.44	1,018	
Took ACT/SAT	0.26	0.44	2,868	

Table 4: Summary statistics for the peer motivation sample

Notes: The table shows means and standard deviations and the number of students observed with each variable for the 2,868 students included in the peer motivation sample.

	Male	Black	Free lunch	Age	Pred. achieve-
	(1)	(2)	(3)	(4)	$\frac{\text{ment}}{(5)}$
Panel A: separate regression	s for each p	peer variabl	е		
Peer motivation	0.002	-0.007	-0.005	-0.023	0.025
	(0.012)	(0.006)	(0.009)	(0.017)	(0.017)
Peer reading achievement	0.017	-0.008	-0.014	-0.024	0.034
	(0.015)	(0.009)	(0.021)	(0.023)	(0.028)
Peer math achievement	0.024	-0.012	-0.028*	-0.020	0.042
	(0.015)	(0.010)	(0.016)	(0.028)	(0.030)
Panel B: joint rearessions fo	r all peer v	ariables			
Peer motivation	0.002	-0.007	-0.004	-0.022	0.024
	(0.012)	(0.006)	(0.010)	(0.016)	(0.017)
Peer reading achievement	-0.000	0.002	0.009	-0.015	0.006
<u> </u>	(0.021)	(0.010)	(0.029)	(0.029)	(0.035)
Peer math achievement	0.024	-0.013	-0.033	-0.009	0.037
	(0.021)	(0.012)	(0.020)	(0.036)	(0.037)
p-value (joint significance)	0.44	0.37	0.22	0.42	0.33
Observations (both panels)	2,861	2,766	2,730	2,845	2,868

Table 5: Balancing tests for peer motivation and peer achievement

Notes: The table shows estimates of regressions of students' socio-demographic characteristics on the characteristics of their classmates. Estimates are based on the peer motivation sample. Predicted achievement in column 5 is constructed from a regression of the averaged reading and math score at the end of students' first year in Project STAR on the four socio-demographic characteristics and school-by-entry-grade fixed effects. In Panel A, each coefficient comes from a separate regression of the outcome indicated in the column header on the peer variable indicated in the row. In Panel B, coefficients are instead based on a single regression in which all peer variables enter jointly. The p-value reported in Panel B comes from an F test for the joint significance of the three peer variables. All regressions in both panels control for school-by-entry-grade fixed effects. Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.

	Re	eading sco	ore	1	Math scor	e
	(1)	(2)	(3)	(4)	(5)	(6)
Peer motivation	0.077***	0.073***	0.071***	0.034	0.029	0.025
	(0.023)	(0.023)	(0.023)	(0.032)	(0.031)	(0.032)
Peer achievement controls	No	Yes	Yes	No	Yes	Yes
Peer demographic controls	No	No	Yes	No	No	Yes
Observations	2,185	$2,\!185$	2,185	$2,\!196$	2,196	$2,\!196$

Table 6: Peer motivation and entry-grade achievement

Notes: The table shows estimates of the effect of peer motivation on achievement in reading (columns 1-3) and math (columns 4-6) at the end of students' first year in Project STAR. Estimates are based on the peer motivation sample. All regressions control for own socio-demographic characteristics, a dummy for small class, and school-by-entry-grade fixed effects. Regressions in columns 2, 3, 5, and 6 additionally control for averages of classmates' reading and math achievement in the previous school year, and regressions in column 3 and 6 additionally control for averages of classmates' socio-demographic characteristics. Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.

	$\begin{array}{c} \text{Motivation score} \\ (1) \end{array}$	Self-concept score (2)
Peer motivation	-0.004	0.001
	(0.028)	(0.027)
Observations	2,276	2,276

Table 7: Peer motivation and own motivation and self-concept

Notes: The table shows estimates of the effect of peer motivation on own motivation and self-concept at the end of students' first year in Project STAR. Regressions control for own socio-demographic characteristics, averages of classmates' reading and math achievement in the previous school year, averages of classmates' socio-demographic characteristics, a dummy for small class, and school-by-entry-grade fixed effects. Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.

	Grade	es 5-8	High	school	College
	reading scores (1)	math scores (2)	GPA (3)	gradua- tion (4)	$\frac{\text{took}}{\text{ACT/SAT}}$ (5)
Peer motivation	-0.023 (0.020)	-0.024 (0.022)	-0.457 (0.422)	-0.031^{*} (0.017)	-0.009 (0.009)
Peer achievement controls Peer demographic controls Observations	Yes Yes 2,118	Yes Yes 2,119	Yes Yes 665	Yes Yes 1,018	Yes Yes 2,868

Table 8: Peer motivation and longer-term educational success

Notes: The table shows estimates of the effect of peer motivation on the outcome variables indicated in the column headers. Regressions control for own socio-demographic characteristics, averages of classmates' reading and math achievement in the previous school year, averages of classmates' socio-demographic characteristics, a dummy for small class, and school-by-entry-grade fixed effects. Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.

- ONLINE APPENDIX -

A Data appendix

In this appendix, I provide additional details about the Project STAR data. The appendix is very similar, and in parts identical, to the data appendix prepared for a previous paper, in which I use data from the same experiment (Bietenbeck, 2020).

Project STAR was planned and implemented by a consortium of researchers from four universities and various state institutions in Tennessee. The experiment ran from the beginning of kindergarten until the end of third grade, but some researchers continued to collect data on participating students in the years afterwards, see Finn et al. (2007) for details. The Project STAR public use file, which is the basis for the empirical analysis in this paper, combines these data such that students can be followed throughout their scholastic careers until the end of high school. In what follows, I present the main independent and dependent variables that I draw from this dataset.

Academic motivation. As described in the main text, students participating in Project STAR were assessed on their academic motivation and self-concept using the Self-Concept and Motivation Inventory (SCAMIN; Milchus, Farrah, and Reitz, 1968) in the spring of each year from kindergarten through third grade. The group-administered, standardized psychological test asks students to indicate pictorially their response to different situations. Based on the answers, a motivation score and a self-concept score are calculated for each student. These scores are included in the public use file.

Tests in personality psychology are often judged by their levels of content-related, construct-related, and criterion validity (Borghans et al., 2008). Content-related validity concerns qualitative judgments by experts about whether a test adequately represents the psychological construct of interest. Construct-related validity refers to the degree to which a test actually measures what it claims to measure and is often assessed using factor analysis. Criterion validity concerns the ability of a test to predict contemporaneous and future outcomes. Finally, another important measure of test quality is reliability, as captured for example by test-retest correlations.

Several previous studies and my own analysis of data from Project STAR indicate a high quality of the SCAMIN early elementary form, which was administered in grades 1-3. Thus, Finn and Cox (1992) point out its strong content validity due to the careful and structured approach taken when creating questions. McIntire and Drummond (1976) show that the motivation score based on the early elementary form correlates with a conceptually related score from the more widely used Coopersmith Self-Esteem Inventory Scales, providing some evidence of construct validity. My results in Section 4 establish criterion validity, as they show that motivation scores predict a wide range of contemporaneous and future outcomes.

Regarding the reliability of the early elementary form, Drummond and McIntire (1975) calculate five-months test-retest correlations of motivation scores of 0.37 and 0.51 in samples of first and second grade students, respectively. Using data from Project STAR, I find a one-year test-retest correlation of 0.31 for both first-grade and second-grade motivation scores. These values are broadly similar to test-retest correlations found for personality traits in children: for example, Measelle et al. (2005) document one-year test-retest correlations for Big Five traits ranging from 0.33 to 0.59 in children aged six to seven, and a meta study by Roberts and DelVecchio (2000) finds an average test-retest correlation of 0.43 for Big Five Traits in children aged six to eleven.

The available evidence paints a different picture of the quality of the SCAMIN preschool/kindergarten form, which was administered in the spring of kindergarten. Thus, Davis, Sellers, and Johnston (1988) analyzed the form's questions using factor analysis and found that they could recover the motivation and self-concept subscales only after disregarding some of the questions, which casts doubt on its construct validity. Moreover, Online Appendix Table A.1 shows that kindergarten motivation scores do not predict any of the measures of educational success studied in the paper, indicating that it has very low (or indeed no) criterion validity.

As for reliability, Davis and Johnston (1987) found three-week test-retest correlations for kindergarten motivation scores of 0.45-0.58 in a sample of 167 kindergarten students. However, Online Appendix Table A.2 shows that kindergarten motivation scores are slightly *negatively* correlated with motivation scores in later grades in the larger sample of Project STAR. As the later scores based on the early elementary form are supposed to measure the same underlying construct (academic motivation), this casts serious doubt on the reliability of the motivation scores based on the preschool/kindergarten form. Given the breadth and severity of these problems, I decided not to use the kindergarten motivation scores in my analysis.

Test scores. At the end of each school year from kindergarten through third grade, students in Project STAR wrote the grade-specific version of the Stanford Achievement Test. From fifth grade through eighth grade, students who were still residing in Tennessee took the Comprehensive Test of Basic Skills (CTBS) as part of a statewide testing program.¹⁵ Both tests are standardized multiple-choice assessments with com-

¹⁵An unrepresentative subsample of students took the CTBS also in fourth grade, see Finn et al. (2007). Due to the selective nature of this subsample, I chose not to analyze fourth-grade test scores.

ponents in reading and math. The second- and third-grade versions of the Stanford Achievement Test further include tests of word study skills and listening skills.

The public use file contains Stanford Achievement Test scores for all students who took these tests. However, it contains CTBS scores only for students who were on grade level, i.e. students who attended grade 5/6/7/8 in 1991/1992/1993/1994, respectively. This implies that test scores are not observed for a number of students who had been retained in grade by those years.¹⁶ Diane Schanzenbach generously provided me with a different version of the Project STAR data, which contains CTBS scores for students who attended grades 5-8 in Tennessee in any year between 1990 and 1997. Test scores are provided as scale scores, which are comparable across grade levels (Finn et al., 2007). In order to increase sample size, I define test scores for a given grade level as scores obtained in the school year in which participating students were supposed to be in that grade (e.g., eighth-grade scores are defined as scores obtained in 1994, even though some students were attending seventh grade in that year).

Classroom behavior. In November 1989, fourth-grade teachers of a subset of former participants in Project STAR were asked to rate their students on their behavior. Specifically, teachers completed a questionnaire that asked them how often each student had engaged in 31 different behaviors over the last two to three months. Ratings were recorded on a scale from 1 ("never") to 5 ("always"), and ratings of 28 of these behaviors were consolidated into four indices. The effort index includes items such as whether a student is persistent when confronted with difficult problems, whether she completes her homework, and whether she gets discouraged easily when encountering an obstacle in schoolwork. The initiative index is based on such items as whether a student participates actively in classroom discussions, whether she does more than just the assigned work, and whether she often asks questions. The discipline index captures such characteristics as whether a student often acts restless, whether she needs reprimanding, and whether she interferes with peers' work. The value index measures how much a student appreciates the school learning environment.¹⁷

During the 1993-94 school year, eighth-grade math and English teachers of a different subset of participants were asked about student behaviors on a similar but shorter

¹⁶Note that students who were retained in grade at any point between kindergarten and third grade dropped out of the STAR cohort and therefore did not write the subsequent Stanford Achievement Tests. However, these students did write the CTBS in later grades as long as they stayed in Tennessee.

¹⁷Note that what the paper refers to as the "discipline index" is the inverse of the "index of nonparticipatory behavior" in the original data. See Finn et al. (2007) for a complete listing of the behaviors included in each of the indices.

questionnaire. Thirteen of these behaviors were again consolidated into four indices measuring each student's effort, initiative, discipline, and value. For my analysis, I averaged the eighth-grade indices across math and English for each student.

High school GPA and graduation. Most students in Project STAR graduated from high school in 1998, and transcripts were gathered from selected high schools in 1999 and 2000. High schools were chosen for data collection based on the likelihood that participants would attend them given the locations of students' last known middle schools. Course grades from transcripts were transferred to a scale from 0-100 if necessary, and separate GPAs for math, science, and foreign languages were computed and are available in the public use file. The empirical analysis in this paper uses overall GPA, defined as the average of the these three subject-specific GPAs, as an outcome variable.

Information on high school graduation was also derived from the transcripts and cross-checked with data from the Tennessee State Department of Education in ambiguous cases. Nevertheless, graduation status could not be determined with certainty for all students. In these cases, the data collectors made a best guess whether a student "probably graduated" or "probably dropped out" based on the available course grades, information on attendance, and additional information from the Tennessee State Department of Education. The variable used in the empirical analysis codes students who graduated, students who probably graduated, and students who received a General Educational Development certificate as graduates, and students who dropped out and students who probably dropped out as dropouts.

College-test taking. ACT/SAT-test taking was recorded by Krueger and Whitmore (2001), who matched all students in Project STAR to the administrative records of the two companies responsible for these tests in 1998. The outcome variable used in the empirical analysis is an indicator that takes value 1 if a student took either of these college entrance exams in 1998 and 0 otherwise.

	Kinder	rgarten	Grade	es 1-3	Grad	es 5-8	High	school	College
	reading score (1)	math score (2)	reading scores (3)	math scores (4)	reading scores (5)	math scores (6)	GPA (7)	gradua- tion (8)	took ACT/SAT (9)
Motivation in KG	0.006 (0.015)	0.002 (0.014)	-0.016 (0.017)	-0.017 (0.017)	-0.017 (0.018)	0.001 (0.017)	0.025 (0.174)	-0.007 (0.009)	0.006 (0.007)
Observations	5,038	5,038	3,716	3,774	(4,051)	(4,049)	2,015	2,456	5,038
<i>Notes:</i> The table shumotivation in kinderg or reduced-price lunch	ows estimate arten. All reg 1. and age. S	s from regre gressions con tandard erro	ssions of the trol for school rs in parenthe	outcome va Lby-entry-gr sses are clust	riables indica ade fixed effe sered bv schoo	ted in the c cts, dummies ol-bv-entrv-g	olumn heade s for male, bl rade. $* p < 0$.	ers on studer ack, and eligi $10^{**} \ v < 0.02$	its' academic bility for free 5, *** $p < 0.01$.

	educational success
-	and
•	motivation
	Andergarten
۲ م	Ч.І:
	Table 7
·-	Appendix
:-	Unline

Online Appendix Table A.2: Correlations between motivation scores in different grades

Motivation	Kindergarten	Grade 1	Grade 2	Grade 3
Kindergarten	1.000			
Grade 1	-0.042	1.000		
Grade 2	-0.056	0.309	1.000	
Grade 3	-0.047	0.220	0.313	1.000

Notes: The table shows correlations between motivation scores in different grades.

B Results from additional analyses



Online Appendix Figure B.1: Randomization check like in Feld and Zoelitz (2017), distribution of p-values

Notes: The figure reports results from a test for random assignment of students to classes similar to the one conducted in Feld and Zölitz (2017). For this test, I ran separate regressions of the variables indicated above the five plots on class dummies for each school-by-entry-grade cell. After each regression, I conducted an F test for the joint significance of the class dummies and collected the p-value. Under random assignment, these p-values should be distributed roughly uniformly. The plots in this figure show the distributions of these p-values for each variable. The red vertical line indicates the p-value of 0.05.

	Grac	les 1-3	Grac	les 5-8	High	school	College
	reading	math scores	reading	math scores	GPA	gradua-	took
	SCOTES		scores			tion	ACT/SAT
	(1)	(2)	(3)	(4)	(5)	(9)	(7)
Panel A: baseline specificat	ion						
Motivation in grades 1-3	0.050^{***}	0.056^{***}	0.054^{***}	0.056^{***}	0.292^{**}	0.008	0.017^{***}
	(0.011)	(0.012)	(0.014)	(0.013)	(0.142)	(0.007)	(0.005)
Observations	8,530	8,678	7,497	7,493	3,360	4,368	9,072
R^2 (within)	0.097	0.067	0.124	0.115	0.109	0.064	0.109
Panel B: specification with	controls for	achievement in	kinderaarten				
Motivation in grades 1-3	0.043^{***}	0.049^{***}	0.049^{***}	0.051^{***}	0.269^{**}	0.008	0.015^{***}
)	(0.010)	(0.010)	(0.012)	(0.012)	(0.133)	(0.001)	(0.004)
Observations	(8,530)	8,678	7,497	7,493	3,360	(4,368)	9,072
R^2 (within)	0.283	0.264	0.239	0.256	0.173	0.069	0.147
$\delta(Rmax = 1.3 \times R^2)$	12.896	17.222	14.174	15.845	14.031	7.536	7.799
$\delta(Rmax = 1.6 \times R^2)$	6.511	8.727	7.187	8.036	7.074	3.793	3.941
Notes: The table quantifies t regressions in Table 2 to zero	he amount o The analysis	f omitted variab	le bias that w net.hod by Ost	yould be needed ⁻ er (2019) and con	to drive the connares the cont	oefficient on m fficient estimate	otivation in the $\frac{1}{2}$ values
from baseline regressions (Par	nel A) with t	hose from regress	sions which ac	Iditionally contro	l for achievem	ent in kinderga	arten (Panel B).
Specifications in both panels the impact of unobservables to	are identical o the impact o	to those in Table of the controls for	e Z. The low r achievement	er rows in Panel in kindergarten t	B show estimation is that would drive	ates of δ , which we the coefficier	h is the ratio of it on motivation
to zero. To compute δ , one n	eeds to make	an assumption a	about the hyp	othetical maximu	um R^2 achieva	ble if all releva	nt controls were
observed, the <i>Rmax</i> . Oster (:	2019) suggest	s setting Rmax	equal to 1.3 t	imes the R^2 from	the controlle	d regression. F	anel B presents
results using this value and th	e more conse: cote as miise	rvative value of I	.b. Calculatio	ns of <i>d</i> are made 2 ² is colouleted u	using the Stat	a package -psa	calc- and treat
errors in parentheses are cluste	ered by schoc	ol-by-entry-grade.	* $p<0.10, **$	p < 0.05, *** p < 0.05	01.	y-ettu y-grade	ntantado (etta

	Grad	les 1-3	Grae	les 5-8	High :	school	College
-	reading scores (1)	math scores (2)	reading scores (3)	math scores (4)	GPA (5)	gradua- tion (6)	took ACT/SAT (7)
Panel A: boys Motivation in grades 1-3	0.022*	0.025*	0.022	0.034**	0.344* 0.107)	0.017*	0.012**
Observations	(0.013) 4,483	(4,560)	(0.010) 3,883	3,878	(0.191) 1,632	2,186	4,781
Panel B: $girls$ Motivation in grades 1-3	0.080***	0.101***	0.102^{***}	0.092***	0.183	0.000	0.022***
Observations	(0.015) 4,047	(0.010) $4,118$	(0.017) 3,614	(0.010) 3,615	(0.201) 1,728	(0.010) $2,182$	(0.008) 4,291
Panel C: black students Motivation in grades 1-3	0.060***	0.088***	0.084***	0.084***	-0.036	0.001	0.022***
Observations	(0.01 <i>1)</i> 3,042	(0.019) 3,046	(0.020) 2,590	(0.022) $2,588$	(0.297) 751	(0.014) 1,317	(0.007) 3,178
Panel D: non-black student Motivation in grades 1-3	5 0.038***	0.033***	0.033**	0.039***	0.389**	0.012	0.014**
Observations	(0.012) 5,488	(0.012) 5,632	(0.014) $4,907$	(0.014) $4,905$	(0.155) 2,609	(0.008) 3,051	(0.000) 5,894
Panel E: students eligible fo Motivation in grades 1-3	or free lunch 0.058***	0.072***	0.068***	0.066***	0.056	-0.002	0.023***
Observations	(0.013) 5,039	(0.015) 5,136	(0.017) $4,406$	(0.016) $4,403$	(0.242) 1,491	(0.010) 2,216	(0.006) 5,397
Panel F: students not eligit Motivation in grades 1-3	ble for free lu 0.022	unch 0.015	0.016	0.030*	0.373^{**}	0.014	-0.001
Observations	(0.015) 3,491	(0.014) $3,542$	(0.018) 3,091	(0.017) 3,090	(0.159) 1,869	(0.009) 2,152	(0.008) 3,675
Notes: The table shows estin motivation, averaged across gr Panel D includes non-black stu eligible for free or reduced-price are clustered by school-by-entr	nates from re- ades 1-3, sep udents and ac e lunch and a v-grade. * v -	egressions of the arately for differed attacks of the arately for differed ditionally studer defitionally stude < 0.10 . ** $p < 0.05$.	outcome vari- ent groups of its with missi nts with missi $^{***} v < 0.01.$	ables indicated in students. Specific ng information or ng information or	the column l ations follow t this variable.	headers on stu he ones in Pan Panel F incluc Standard error	lents' academic el B of Table 2. les students not s in parentheses

Online Appendix Table B.2: Own motivation and educational success, heterogeneity by student characteristics

	Grae	les 1-3	Grae	les 5-8	High	school	College
	reading	math scores	reading	math scores	GPA	gradua-	took
	(1)	(2)	(3)	(4)	(5)	(6)	AC1/SA1 (7)
Panel A: sample restrictea	<i>to students</i>	observed with ac	hievement in	ı kindergarten			
Motivation in grades 1-3	0.025^{**}	0.032^{**}	0.032^{*}	0.039^{**}	0.169	0.005	0.020^{***}
	(0.012)	(0.013)	(0.018)	(0.016)	(0.159)	(0.008)	(0.007)
Observations	4,099	4,179	3,666	3,662	2,053	2,431	4,219
Panel B: specification with	<i>v</i> control for	academic self-con	ncept in graa	les 1-3			
Motivation in grades 1-3	0.038^{***}	0.045^{***}	0.048^{***}	0.050^{***}	0.181	0.005	0.014^{***}
	(0.010)	(0.011)	(0.013)	(0.012)	(0.134)	(0.007)	(0.005)
Observations	8,530	8,678	7,497	7,493	3,360	4,368	9,072
Notes: The table shows estin	mates from re	egressions of the	outcome vari	ables indicated in	a the column	neaders on stu	dents' academic
motivation, averaged across g	grades 1-3. Al	l specifications ar	e variations o	of the regressions	in Panel B of	Table 2. Pane	l A restricts the
sample to students who are ob	served with k	indergarten readir	ng or math sco	res. Panel B add	s a control for s	tudents' acader	mic self-concept,
averaged cross grades $1-3$. St _i	andard errors	in parentheses are	e clustered by	school-by-entry-	grade. * $p < 0.1$	0, ** p<0.05, *	p < 0.01.

Online Appendix Table B.3: Own motivation and educational success, robustness checks

	Male	Black	Free lunch	Age	Pred. achieve- ment
	(1)	(2)	(3)	(4)	(5)
p-value Observations	.14 2,861	.99 2,766	.26 2,730	$.30 \\ 2,845$.69 2,868

Online Appendix Table B.4: Randomization check like in Chetty et al. (2011)

Notes: The table reports results from a test for random assignment of students to classes similar to the one conducted in Chetty et al. (2011). The intuition of this test is that if students were indeed randomly assigned to classes, then class dummies should not predict their predetermined characteristics. For this table, I regressed each of the variables indicated in the column headers on school-by-entry-grade fixed effects and class dummies (leaving out one dummy per school-by-entry-grade cell to avoid collinearity). I then conducted an F test for the joint significance of all class dummies. The table reports the corresponding p-values.

	No. of tests	No. of	p-values	below	Share o	f p-value	s below
		10%	5%	1%	10%	5%	1%
Male	145	16	9	3	11.03%	6.21%	2.07%
Black	63	4	2	1	6.35%	3.17%	1.59%
Free lunch	121	11	6	2	9.10%	4.96%	1.65%
Age	147	12	6	3	8.16%	4.08%	2.04%
Pred. achievement	147	11	4	1	7.48%	2.72%	0.68%

Online Appendix Table B.5: Randomization check like in Feld and Zoelitz (2017), number of p-values below certain thresholds

Notes: The table reports results from a test for random assignment of students to classes similar to the one conducted in Feld and Zölitz (2017). For this test, I ran separate regressions of the variables indicated in rows on class dummies for each school-by-entry-grade cell. After each regression, I conducted an F test for the joint significance of the class dummies and collected the p-value. Under random assignment, the shares of p-values below certain confidence levels should be close to this this level (for example, about five percent of p-values should be below 0.05). The table shows the number of tests conducted for each variable and the number and share of p-values below the thresholds of 10%, 5% and 1%. The number of tests conducted is lower than the number of school-by-entry-grade cells, 147, for some variables due to missing data or due to collinearity (for example, if all students entering a certain school in a certain grade were black).

	Readin	g score	Math	score
	(1)	(2)	(3)	(4)
Peer motivation	0.077^{***}	0.071^{***}	0.034	0.025
	(0.022)	(0.023)	(0.032)	(0.032)
Peer achievement controls	No	Yes	No	Yes
Peer demographic controls	No	Yes	No	Yes
Observations	2,185	$2,\!185$	$2,\!196$	2,196
R^2 (within)	0.112	0.123	0.271	0.282
$\delta(Rmax = 1.3 \times R^2)$		2.458		0.132
$\delta(Rmax = 1.6 \times R^2)$		1.282		0.103

Online Appendix Table B.6: Peer motivation and entry-grade achievement, analysis of omitted variable bias

Notes: The table quantifies the amount of omitted variable bias that would be needed to drive the coefficient on peer motivation in the regressions in Table 6 to zero. The analysis is based on the method by Oster (2019) and compares the coefficient estimates and R^2 values from baseline regressions (columns 1 and 3) with those from regressions which additionally control for averages of classmates' reading and math achievement in the previous school year and averages of classmates' socio-demographic characteristics (columns 2 and 4). For further details on controls included in the specifications, see Table 6. The last two rows in the table show estimates of δ , which is the ratio of the impact of unobservables to the impact of the controls for peer achievement and socio-demographic characteristics that would drive the coefficient on peer motivation to zero. To compute δ , one needs to make an assumption about the hypothetical maximum R^2 achievable if all relevant controls were observed, the Rmax. Oster (2019) suggests setting Rmax equal to 1.3 times the R^2 from the controlled regression. The table presents results using this value and the more conservative value of 1.6. Calculations of δ are made using the Stata package -psacalc- and treat school-by-entry-grade fixed effects as nuisance parameters (that is, the R^2 is calculated within school-by-entry-grade cells). Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.

	Reading score	Math score
	(1)	(2)
Panel A: boys		
Peer motivation	0.095***	0.036
	(0.030)	(0.042)
Observations	1,207	1,220
Panel B: girls		
Peer motivation	0.047	0.007
	(0.044)	(0.042)
Observations	978	976
Panel C: black stude	nts	
Peer motivation	0.084^{***}	0.070
	(0.027)	(0.057)
Observations	962	956
Panel D: non-black s	tudents	
Peer motivation	0.067**	-0.006
	(0.032)	(0.034)
Observations	1,223	1,240
Panel E: students eli	gible for free lunch	
Peer motivation	0.073**	0.034
	(0.031)	(0.045)
Observations	1,372	1,374
Panel F: students no	t eligible for free lunch	
Peer motivation	0.064	0.015
	(0.043)	(0.037)
Observations	813	822

Online Appendix Table B.7: Peer motivation and entry-grade achievement, heterogeneity by student characteristics

Notes: The table shows estimates of the effect of peer motivation on achievement in reading and math separately for different groups of students. All regressions control for own socio-demographic characteristics, averages of classmates' reading and math achievement in the previous school year, averages of classmates' socio-demographic characteristics, a dummy for small class, and school-by-entry-grade fixed effects. Panel D includes non-black students and additionally students with missing information on this variable. Panel F includes students not eligible for free or reduced-price lunch and additionally students with missing information on this variable. Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.

	Interaction	with small class	Regular-size	d classes only
	reading	math score	reading	math score
	score		score	
	(1)	(2)	(3)	(4)
Peer motivation	0.094***	0.038	0.091***	0.048
	(0.029)	(0.044)	(0.029)	(0.050)
\times small class	-0.071	-0.041		
	(0.052)	(0.066)		
Small class	0.065	0.092		
	(0.051)	(0.062)		
Peer achievement controls	Yes	Yes	Yes	Yes
Peer demographic controls	Yes	Yes	Yes	Yes
Observations	$2,\!185$	2,196	$1,\!663$	$1,\!671$

Online Appendix Table B.8: Peer motivation and entry-grade achievement, heterogeneity by class size

Notes: The table shows estimates of the effect of peer motivation on achievement in reading and math. In columns 1 and 2, peer motivation is interacted with the small-class dummy. In columns 3 and 4, the sample is restricted to students in regular-sized classes. Regressions control for own socio-demographic characteristics, averages of classmates' reading and math achievement in the previous school year, averages of classmates' socio-demographic characteristics, a dummy for small class, and school-by-entry-grade fixed effects. Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.

	$\begin{array}{c} \text{Reading score} \\ (1) \end{array}$	Math score (2)
Share of peers with top 33% motivation	0.148	0.059
	(0.189)	(0.292)
Share of peers with bottom 33% motivation	-0.412^{**}	-0.221
	(0.159)	(0.177)
Peer achievement controls	Yes	Yes
Peer demographic controls	Yes	Yes
Observations	$2,\!185$	2,196

Online Appendix Table B.9: Peer motivation and entry-grade achievement, bad apples and shining lights

Notes: The table shows estimates of the effect of peer motivation on achievement in reading and math. Peer motivation is measured as the shares of classmates with top 33% and bottom 33% motivation scores. Regressions control for own socio-demographic characteristics, averages of classmates' reading and math achievement in the previous school year, averages of classmates' socio-demographic characteristics, a dummy for small class, and school-by-entry-grade fixed effects. Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.

	word study skills score (1)	listening score (2)
Peer motivation	0.082^{***}	0.025
Peer achievement controls	Yes	Yes
Peer demographic controls Observations	Yes 2,507	Yes 2,187

Online Appendix Table B.10: Peer motivation and entry-grade achievement in other subjects

Notes: The table shows estimates of the effect of peer motivation on achievement in word study skills and listening, which were assessed by the Stanford Achievement Test next to reading and math. Achievement scores are standardized to have mean 0 and SD 1 in each subject. Regressions control for own socio-demographic characteristics, averages of classmates' reading and math achievement in the previous school year, averages of classmates' socio-demographic characteristics, a dummy for small class, and school-by-entry-grade fixed effects. Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.

		Entry	grade		Grad	es 5-8	High :	school	College
	reading score	math score	word st. skills	listening score	reading scores	math scores	GPA	gradua- tion	took ACT/SAT
	(1)	(2)	score (3)	(4)	(5)	(9)	(2)	(8)	(6)
Peer motivation	0.071	0.025	0.082	0.025	-0.023	-0.024	-0.457	-0.031	-0.009
	[0.003]	[0.438]	[0.001]	[0.366]	[0.259]	[0.275]	[0.280]	[0.075]	[0.342]
	[0.040]	[0.785]	[0.012]	[0.785]	[0.785]	[0.785]	[0.757]	[0.458]	[0.785]
Peer ach. controls	\mathbf{Yes}	\mathbf{Yes}	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes	\mathbf{Yes}	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$
Peer dem. controls	\mathbf{Yes}	Yes	\mathbf{Yes}	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	\mathbf{Yes}	\mathbf{Yes}	Yes	$\mathbf{Y}_{\mathbf{es}}$
Observations	2,185	2,196	2,507	2,187	2,118	2,119	665	1,018	2,868
Notes: The table shows e sets of p-values. The p-va Appendix Table B.10. Th and Wolf (2005a,b). To ii	stimates of th dues in bracke te p-values in j mplement this	ne effect of pe ets shown dire italics and bre s procedure, I	ser motivation setly below the ackets in the ne use the Stata	on the outcon coefficient est ext row are cor rwolf comma	ne variables in imates are ba rected for mul und described	dicated in the sed on the mai tiple hypothes in Clarke, Ror	column head n estimates ir is testing usin nano, and Wc	ers along with 1 Tables 6 and g the procedu df (2019).	a two different 8 and Online re by Romano

Online Appendix Table B.11: Peer motivation and educational success, correction for multiple hypothesis testing

4			ni ne si emos	iod ror for hai	na observad v	with	
	entry	r grade	grade	s 5-8	high high	school	college
	reading score (1)	math score (2)	reading scores (3)	math scores (4)	GPA (5)	graduation (6)	took ACT/SAT (7)
Peer motivation	0.008	0.005	-0.015*	-0.015^{*}	-0.012	-0.007	
Peer achievement controls	(euuu) Yes	Yes	Yes	Yes	\mathbf{Yes}	Yes	
Peer demographic controls	${ m Yes}_{ m 0.060}$	Yes	Yes	Yes	Yes	Yes	I
Observations	2,868	2,868	2,868	2,868	2,868	2,868	
Notes: The table shows estima	tes from regr is empty her	essions of dummination of dummination of dummination of the second secon	ies for being ol	bserved with th bserved for all	ne outcomes ir students Rec	idicated in the c	olumn headers for own socio-
demographic characteristics, av	erages of clas	smates' reading a	and math achie	evement in the	previous scho	ol year, averages	of classmates'
socio-demographic characteristi	ics, a dummy	for small class, <i>e</i>	and school-by-e	entry-grade fixe	ed effects. Star	ndard errors in p	arentheses are
clustered by school-by-entry-gra	ade. * $p < 0.10$, ** p<0.05, *** 1	p < 0.01.				

at
selective
of
analysis
success,
educational
and
motivation
Peer
le B.12:
Tab
Appendix
Online

	Reading score	Math score		
	(1)	(2)		
Panel A: more than 50% of peers observed with motivation scores				
Peer motivation	0.061^{*}	0.020		
	(0.031)	(0.034)		
Peer achievement controls	Yes	Yes		
Peer demographic controls	Yes	Yes		
Observations	1,590	$1,\!602$		
Panel B: more than 66% of peers	s observed with motivati	on scores		
Peer motivation	0.061^{*}	0.026		
	(0.034)	(0.038)		
Peer achievement controls	Yes	Yes		
Peer demographic controls	Yes	Yes		
Observations	1,094	1,104		
Panel C: more than 75% of peers	s observed with motivati	on scores		
Peer motivation	0.071	0.042		
	(0.056)	(0.052)		
Peer achievement controls	Yes	Yes		
Peer demographic controls	Yes	Yes		
Observations	643	647		

Online Appendix Table B.13: Peer motivation and entry-grade achievement, results for subsamples with information on motivation for a high share of peers

Notes: The table shows estimates of the effect of peer motivation on achievement in reading and math. In Panel A/B/C, the sample is restricted to students for whom more than 50/66/75percent of their classmates are observed with motivation scores from the previous school year. Regressions control for own socio-demographic characteristics, averages of classmates' reading and math achievement in the previous school year, averages of classmates' socio-demographic characteristics, a dummy for small class, and school-by-entry-grade fixed effects. Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.

	$\begin{array}{c} \text{Reading score} \\ (1) \end{array}$	Math score (2)
Peer motivation	0.068^{***}	0.011
Peer achievement controls	(0.024) Yes	(0.051) Yes
Peer demographic controls	Yes	Yes
Observations	2,185	2,196

Online Appendix Table B.14: Peer motivation and entry-grade achievement, controlling for peer self-concept

Notes: The table shows estimates of the effect of peer motivation on achievement in reading and math. Regressions control for own socio-demographic characteristics, averages of classmates' socio-demographic characteristics and their math and reading scores in the previous school year, a dummy for small class, and school-by-entry-grade fixed effects. Regressions also control for the average of classmates' self-concept score in the previous school year. Standard errors in parentheses are clustered by school-by-entry-grade. * p < 0.10, ** p < 0.05, *** p < 0.01.